Reviewer #1:

Specific Comments to Authors:

answer: 239-240, Table 2 MN and NMN

EDITORIAL OFFICE’S COMMENTS

(1) Science editor: 147-150.

(2) Company editor-in-chief: original figure documents 217-218, 224-225
Response to Reviewers' comments

Dear Editor,

We thank you for your careful consideration of our manuscript. We appreciate your response and overall positive initial feedback and made modifications to improve the manuscript. After carefully reviewing the comments made by the Reviewers, we made a point-to-point response and retained traces of revision in the article, please refer to the article.
We hope that you will find the revised paper suitable for publication, and we look forward to contributing to your journal. Please do not hesitate to contact us with other questions or concerns regarding the manuscript.

Best regards,

Reviewer #1
Thank you for the corrected manuscript. There are very few comments. In table 2, next to postoperative icon, there is 72. What do you mean? reference? No need for 1 mmHg= 0.133 kPa in the history part. There is no metanephrine value written in laboratory investigations, which is 31.9 as mentioned in the table. Please capitalize letter B in Brain in imaging examination part. In the treatment part, there is olmesartan/hydrochlorothiazide tablet, it is a combination. Please write each concentration. Remove the word of, it is the last word in conclusion part. Connect the word with in figure 1 legend. Thank you.

Comment 1: In table 2, next to postoperative icon, there is 72. What do you mean? reference?
Response: No need for 72, 72 can be deleted.

Comment 2: No need for 1mmHg= 0.133 kPa in the history part.
Response: 1mmHg= 0.133 kPa has been deleted.

Comment 3: There is no metanephrine value written in laboratory investigations, which is 31.9 as mentioned in the table.
Response: The laboratory part of the article has supplemented that the value of Metanephrine (MN) was 31.9ng/L(reference range < 96.60 ng/L).

Comment 4: Please capitalize letter B in Brain in imaging examination part.
Response: In the imaging section, the letter B in the brain has been adjusted to capital B.

*Comment 5:* In the treatment part, there is olmesartan/hydrochlorothiazide tablet, it is a combination. Please write each concentration.
Response: In the treatment part, the drug is a single compound fixed preparation of antihypertensive drugs, including olmesartan axetil tablets 20mg and hydrochlorothiazide tablets 12.5mg.

*Comment 6:* Remove the word of, it is the last word in conclusion part.
Response: The last word of the conclusion section has been deleted.

*Comment 7:* Connect the word with in figure 1 legand.
Response: Figure 1 has been revised again, connect in Figure 1 has been deleted